Author's response to reviews

Title: How risky is caring for emergency patients at risk of malpractice litigation: a population based epidemiological study of Taiwan's experiences

Authors:

   Che-Ming Yang (cyang@tmu.edu.tw)
   Shin-Han Tsai (shtsai@tmu.edu.tw)
   Wen-Ta Chiu (wtchiu@tmu.edu.tw)

Version: 4 Date: 15 September 2009

Author's response to reviews:

Dear Editor:

Thank you for your mail dated 14 Sep 2009. We have made the following revisions per your instruction.
1. Full citations still cannot be found in Pubmed for Reference 2 & 11.
2. Qualifications have been removed from the manuscript and submissions system
3. Keywords have been removed from the abstract.
4. Weblinks have been added to Reference 9/18/19.
5. The blank page in the end has been deleted.

Thank you.

Sincerely,

Cheming Yang